
\documentclass[border=5pt,multi=true]{standalone}
\usepackage{standalone}
\usepackage{ifthen}
\usepackage{microtype}
\usepackage{lineno}
\usepackage{xspace}
\usepackage{graphicx}
\usepackage{color}
\usepackage{colortbl}
\usepackage{rotating}
\usepackage{subfigure}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{upgreek}
\usepackage{hyperref}
\usepackage{hypcap}
\usepackage{cancel}
\usepackage{tabularx}
\usepackage{array}
\usepackage{multirow}
\usepackage{bbm}
\usepackage{cite}
\usepackage{mciteplus}

\begin{document}
  
    
    
    \resizebox{0.99\textwidth}{!}{
    \begin{tabular}{|c|c|c|c|c|}
\hline
 Observable & Measurement  & Source  & SM prediction &  References  \\
\hline
\multicolumn{5}{|c|}{ $ B\xspace\xspace^0_ s\xspace\xspace$ \xspace system} \\
\hline
$\Delta m_s$ ($ {\rm \,ps^{-1}}\xspace$)  & $ 17.719 \pm 0.043 $  & HFAG 2012~\cite{HFAG} & $17.3 \pm 2.6 $ & \cite{Lenz:2011ti,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
                         & $ 17.725 \pm 0.041 \pm 0.026  $  & LHCb ($0.34 \mbox{\,fb}^{-1}\xspace$)~\cite{LHCb-CONF-2011-050} &                 &    \\
\hline
$\Delta \Gamma_s$ ($ {\rm \,ps^{-1}}\xspace$)  & $ 0.105 \pm 0.015 $  & HFAG 2012~\cite{HFAG}& $0.087 \pm 0.021 $ & \cite{Lenz:2011ti,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
                         & $ 0.116 \pm 0.018 \pm 0.006  $  & LHCb ($1.0 \mbox{\,fb}^{-1}\xspace$)~\cite{LHCb-CONF-2012-002} &                 &    \\
\hline
$\phi_s$  (rad) &  $-0.044 ^{+0.090}_{-0.085}$ &  HFAG 2012~\cite{HFAG}  &  $-0.036 \pm 0.002$ & \cite{Charles:2011va,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az} \\
               &  $-0.002 \pm 0.083 \pm 0.027$ &  LHCb ($1.0 \mbox{\,fb}^{-1}\xspace$)~\cite{LHCb-CONF-2012-002}  &  &  \\
\hline
$a^s_{\rm sl}$ ($ 10 ^{-4}$)   &  $-17\pm 91\,^{+14}_{-15}$ &  \mbox{D0}\xspace (no $A^b_{\rm SL}$)~\cite{Abazov:2009wg}  & $0.29^{+0.09}_{-0.08}$ & \cite{Charles:2011va,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
                           &  $-105\pm 64$ &  HFAG 2012 (including $A^b_{\rm SL}$ )~\cite{HFAG}  &   & \\
\hline
\multicolumn{5}{|c|}{Admixture of  $ B\xspace\xspace^0$ \xspace and  $ B\xspace\xspace^0_ s\xspace\xspace$ \xspace systems} \\
\hline
$A^b_{\rm SL}$ ($ 10 ^{-4}$)   &  $-78.7\pm 17.1 \pm 9.3$ &  \mbox{D0}\xspace~\cite{Abazov:2011yk}  & $-2.0 \pm 0.3 $ & \cite{Lenz:2011ti,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
\hline
\multicolumn{5}{|c|}{ $ B\xspace\xspace^0$ \xspace system} \\
\hline
$\Delta m_d$ ($ {\rm \,ps^{-1}}\xspace$)  & $ 0.507 \pm 0.004 $  & HFAG 2012~\cite{HFAG}& $0.543\pm 0.091 $ & \cite{Lenz:2012mb,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
\hline
$\Delta \Gamma_d / \Gamma_d$   & $ 0.015 \pm 0.018 $  & HFAG 2012~\cite{HFAG}& $0.0042 \pm 0.0008 $ & \cite{Lenz:2011ti,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
\hline
$\sin2\beta$ &  $0.679 \pm 0.020$ &  HFAG 2012~\cite{HFAG}  & $0.832 \,^{+0.013}_{-0.033} $ & \cite{Charles:2011va,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
\hline
$a^d_{\rm sl}$ ($ 10 ^{-4}$)   &  $-5\pm 56$ &  HFAG 2012~\cite{HFAG}  & $-6.5\,^{+1.9}_{-1.7}$ & \cite{Charles:2011va,Lenz:2006hd,Beneke:1998sy,*Ciuchini:2003ww,*Beneke:1996gn,*Beneke:2003az}\\
\hline
    \end{tabular}
  }
\label{tab:mixing_status}
  
\end{document}
